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19 January 2010
MEMORANDUM FOR CENWP-OP, ATTN: Operations Biologist
SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for Week 03 of 2010, which covers the period from 10 through 16 January 2010.
1 OPERATION SUMMARY

1.1 Daily average river flows ranged from 114.7 to 135.2 kcfs.  Daily average project tailwater ranged from 11.7’ to 13.3’ msl.  Daily average powerhouse forebay elevation ranged from 75.9’ to 77.0’ msl.  Daily average river temperature ranged from 40 to 41ºF.  Secchi disk measurements ranged from 6.0’ to 7.0’.
1.2 Spill Bays 1, which provides attraction flow for Cascades Island, is open 6” during daylight hours.  Daily average spill ranged from 0.5 to 0.6 kcfs.
1.3 Unit Operation.  Powerhouse Two is the priority powerhouse.
1.3.1 Unit 7 has remained out of service since 04 June 2007 for turbine rehab.
1.3.2 Unit 9 has remained out of service since 04 August 2008 for turbine rehab.
1.3.3 Unit 11 has remained out of service since 12 June 2008 due to rotor damage.
1.4 Second Powerhouse fish units provide attraction flow for the Washington Shore (WS) fish ladder.  The units may be placed on standby, as necessary, during nighttime hours to float debris.
1.4.1 F1 remains out of service for annual maintenance.
1.5 Gatewell drawdowns are measured on Wednesdays.
2 ACTIVITIES

2.1 Auxiliary Water System Closures.  The following AWS valves were closed this reporting week: 

2.1.1 FV1-1, 3-7, 3-9, 4-3, and 4-4 remain closed for winter maintenance in Bradford Island.
2.2 STS/VBS inspections.  The following STS/VBS inspections and removal occurred this reporting week.
Table 1.  STS and VBS inspections.  
	Unit
	Date
	Run Hours
	Comments

	11
	NA
	OOS
	STSs removed on 28 August 2008.

	12-18
	NA
	
	STSs scheduled for installation the week of 21 February.


2.3 Dewatering and Fish Salvage.  The following dewatering/fish salvage activities occurred this week:

2.3.1 On 11 January, Fisheries removed four steelhead and approx. 65 sturgeon from B-branch.
2.3.2 On 12 January, Fisheries removed approx. 180 sturgeon from B-branch.
3 RESEARCH  

3.1 FFU- sea lion.  Observations began on 08 January.  Weekly reports of sea lion activity will be available from Fisheries Field Unit starting in February. If you would like to be included on the distribution list contact Robert Stansell-541-374-8801.
3.2 NOAA- PIT tag detectors.  Vibration sensor remains installed on Spill bay 6. 

3.3 ODFW/WDFW sea lions.  Moved the forebay sea lion trap on 19 November.
3.4 Smith-Root sea lion array.  Array remains installed in the UMT.
3.5 WDFW Fish Counting.  Video counting at Washington Shore continues.  
4 FISHWAYS 

4.1 Project biologists inspected 10, 11, 13-15 January.
4.2  Fish Passage Plan Observations.
Table 2.  FPP Out of Criteria items.
	DATE
	LOCATion
	FPP itEMS
	CAUSE
	RESPONSE
	TR#

	BRADFORD ISLAND (BI)- fishway currently dewatered

	CASCADES ISLAND (CI)- nothing to report

	WASHINGTON SHORE (WS) – nothing to report


4.2.1 PH1 PCC diffusers FG2-22A and FG2-19 are closed or partially open.  Repairs scheduled for 2009/2010 in water work period.  

4.2.2 Design of a mechanical velocity meter at the PH2 south monolith is in progress.  

4.3 Fishway Temperatures.  Temperatures are taken 01 March through 30 November.

4.4 Zebra Mussel Monitoring.  No signs of colonization were observed this reporting week.  
4.5 Adult Fishways. 
4.5.1 Calibration was checked on 13 January.
4.5.2 The PH1 ice and trash sluiceway (ITS) remains dewatered.
4.5.2.1 The wooden bulkhead is scheduled for replacement on 22 January.
4.5.3 The Washington Shore fishway entrances remain in manual due to SA-24 board calibration issues.  
4.5.3.1 NUE remains closed.  The other gates are adjusted as needed.
4.5.4 Stab plates remain installed in 11 of 12 FOGs.  One gate remains out for maintenance, with a bulkhead installed.
4.5.5 SLEDs remain installed at PH2.

4.6 Juvenile Fishways.
4.6.1 The B2CC remains closed.
4.6.2 DSM2 remains dewatered.
4.7 Lamprey Fishways.  All LPSs have been dewatered.
4.7.1 The Cascades Is. entrance weir and slot filler remain installed.  
4.7.2 The PH2 NDE ramp remains in the lowered position.
5 CONSTRUCTION
5.1 PH1 Gantry 4.  Crane rails replacement continues.  New crane in FY11.  
5.2 PH1 Ice and Trash Sluiceway Wall Removal.  Wall removal continues.  Outfall removal has been postponed.
5.3 TIE crane.  Load testing occurred the week of 29 November.
6 HAZMAT, SPILLS AND CLEANUP. 
6.1 At 1136 on 13 January, an oil sheen was reported in the parking lot of the Bradford Island Service Building (NRC #928485, OERS #2010-0097, WEMD #10-0128).  A small amount of oil was released from a secondary containment for a hydraulic lift system on a Project truck during a torrential downpour.  An oil sheen flowed from the parking lot into the PH1 forebay via a boat ramp and potentially a roadway drain.  A containment boom was placed downstream and absorbent boom and pads were used.  Total volume estimated was less than one quart.
6.2 An oil sheen was reported in the PH2 forebay near the B2CC around 1500 on 15 January (NRC #928729, OERC #2010-0111, WEMD #10-0153). A small amount of oil was released into the road trench drain between the TIE crane rails at PH2, which feeds into a manhole and then dumps to the forebay upstream of B2CC.  A 3x13 foot sheen appeared in the forebay.  A containment boom and absorbent boom was placed.  Total volume was estimated at less than one cup.
7 TELETYPES AND OTHER NOTEWORTHY ITEMS
7.1 A teletype sent on 16 July 2009, changes PH2 unit priority to: 12,18,15,17,14,13,16.  This change is effective from 1100 on 16 July 2009 until U11 returns to service.
7.2 A teletype sent on 08 January requests BON forebay maintain 71.5’ to 77.0’ msl as a soft constraint, and 71.5’ to 77.5’ msl as a hard constraint from 0001 on 09 January until 1800 on 31 March.  This is a modification of the 01 September teletype.  This request is to reduce risk of flooding machinery and twisting bollards at BON locks, while still supporting spillwall construction at TDA.
7.3 A teletype sent on 08 September requests that BON operate the single highest priority unit at PH1 (currently Unit 1) as needed.  This is to reduce the risk of BON going outside the 1% efficiency range while supporting the high forebay request for TDA spillwall work.  This request is effective immediately until further notice.
7.4 A teletype sent on 30 December requests BON tailwater elevation be maintained at no less than 11.5’ msl during all hours.  This is effective from 1600 on 30 December until further notice.  This request is to provide necessary flows for chum egg incubation.

7.5 A teletype sent on 15 January requests a maximum generation of two fish units and two main units at BON PH2 from 0900 until 1500 on 20 and 21 January.  This request is to provide low flows for diver safety during the BGS inspection dive in the PH2 forebay.

JERRY CARROLL
Acting Operations Manager
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